
Stacking
○ Protons on target

� 6.7e12 and jumped up to 7.0e12 at 10 turns
� 7.35e12 at 11 turns

○ Peak one hour stack rates over the weekend were

� 23.79, 23.49, and 24.79mA/hr
� 24.79mA appears to be artificially high.

○ Average productions were

� 19.39, 16.24, and 15.9 e-6/proton
� Production efficiencies had fallen off.  

□ At least part of this can be attributed to DRF1-4 tripping.
○ Had a peak stack of 160mA on Saturday evening, and emittances need some extra 

attention.
� Ran the flusher and tweaked the ARF2 frequency sweep and voltage
� The horizontal tune was a little high.
� Core cooling was on a little too hard.

○ DRF1-4 tripped off.  

� Experts are looking into this.
� The DRF1 fanback voltage dropped to 4.3MV with this station off.
� We see the effects on Debuncher efficiency when the fanback voltage is less 

than 5MV.
○ Our Debuncher injection bunch train length parameter D:FLXBTL stopped working.  

� DCE11 was rebooted (FLXCAP runs off of that), but that did not cure the 
problem.

� It turns out the Flux Capacitor scope was updating but not talking on the 
network. 

� The scope was power cycled, and calculated parameters started updating.
� The scope had to be cycled at AP10 because the new remote power cycling 

parameter D:SCP10R (P38 MISC <15>) was not working.  This problem was 
found a problem with PLC04, which was power cycled to bring it back to life.

○ BPI10D (circulating beam intensity in the Debuncher) also stopped updating.  The 
python startup script was run and updates resumed. 

○ Pbar Crate 71 slot 1, MADC had a NO Q error.  This MADC is responsible for various 
target station readbacks.  The MADC was reset from D21, and readbacks resumed.
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target station readbacks.  The MADC was reset from D21, and readbacks resumed.

○

Transfers
○ Unstacked 1,279mA in 100 transfers over 31 sets.

� Ave Accumulator to MI efficiency was 97%
� Ave Accumulator to RR efficiency was 90%

□ Some sub-90% transfer efficiencies were caused by dp/p in Recycler.
○ On Transfer 7035 had some controls issues.

� The state device V:APSMOD had a change sent from stacking (mode 7) to 
transfers (mode 12).  The change was sent, but did not go out.

� No beam was lost on this missed transfer since ARF4 LLRF never turned on, 
since the state 12 didn't go out.

� Experts added a check in the sequencer to prevent this from happening again.
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Studies
○ One shot reverse protons during stacking

Requests
○ Debuncher Cooling measurements

� Study requires two hours in stacking mode with beam circulating in the 
Debuncher (no stacking).

� The study requires running a sequencer aggregate that will load appropriate 
TLGs for the study.  Non-stacking events may be taken out of the timelines 
temporarily when this is run.
Studiers are available today.

   Morning Summaries Page 3    



temporarily when this is run.
� Studiers are available today.
� Ralph Pasquinelli, Steve Werkema, and Dave Vander Meulen will be the 

studiers.
� Run Co approved this study after successful completion of the HEP shot.

○ Debuncher gain ramping.

� Two hour study to make changes to the Debuncher momentum cooling gain 
ramps.   

� This is the first of a number of iterations of this study as we try to find a more 
optimal configuration for the Debuncher momentum gain ramps.

� As changing momentum cooling could result in periodic trips of the TWTs and 
power changes to compensate, we expect on the order of a 10% hit to 
stacking.

� We want stable beam for this study (intensity and TLG).
� Brian Drendel, Valeri Lebedev and Vladimir Nagaslaev will be the studiers.

Other Notes
○ Paul's Numbers

� Saturday, 2-2
□ Most in an hour: 23.79 mA at Sat Feb 02 03:49:04 CST 2008 
□ Best: 25.19 mA on 30-Jan-08 
□ Average Production 19.39 e-6/proton Best: 25.41 e-6/proton on 

01/30/2008 
□ Average Protons on Target 6.39 e12 Best: 8.77 e12 on 07/24/2007 
□ Largest Stack .00 mA Best: 271.01 mA on 11/14/2007 

� Sunday, 2-3
□ Most in an hour: 23.49 mA at Sat Feb 02 07:00:20 CST 2008 
□ Best: 25.19 mA on 30-Jan-08 
□ Average Production 16.24 e-6/proton Best: 25.41 e-6/proton on 

01/30/2008 
□ Average Protons on Target 7.00 e12 Best: 8.77 e12 on 07/24/2007 
□ Largest Stack 160.29 mA Best: 271.01 mA on 11/14/2007 

� Monday, 2-4
□ Most in an hour: 24.79 mA at Mon Feb 04 03:19:07 CST 2008 
□ Best: 25.19 mA on 30-Jan-08 
□ Average Production 15.90 e-6/proton Best: 25.41 e-6/proton on 

01/30/2008 
□ Average Protons on Target 7.06 e12 Best: 8.77 e12 on 07/24/2007 
□ Largest Stack 97.80 mA Best: 271.01 mA on 11/14/2007

○ Al's Numbers

� Stacking
□ Pbars stacked: 1274.67 E10
□ Time stacking: 66.95 Hr
□ Average stacking rate: 19.04 E10/Hr

� Uptime
□ Number of pulses while in stacking mode: 98919
□ Number of pulses with beam: 94347
□ Fraction of up pulses was: 95.38%

� The uptime's effect on the stacking numbers
□ Corrected time stacking: 63.85 Hr
□ Possible average stacking rate: 19.96 E10/Hr

� Recycler Transfers
□ Pbars sent to the Recycler: 1274.72 E10
□ Number of transfers : 100
□ Number of transfer sets: 30
□ Average Number of transfer per set: 3.33
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□ Average Number of transfer per set: 3.33
□ Time taken to shoot: 04.55 Hr
□ Time per set of transfers: 09.11 min
□ Transfer efficiency: 94.14%

� Other Info
□ Average POT : 7.13 E12
□ Average production: 18.94 pbars/E6 protons
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